Author's response to reviews

Title: How healthy are chronically ill patients after eight years of homeopathic treatment? - Results from a long term observational study

Authors:

Claudia M Witt (claudia.witt@charite.de)
Rainer Lüdtke (r.luedtke@carstens-stiftung.de)
Nils Mengler (nilsmengler@web.de)
Stefan N Willich (stefan.willich@charite.de)

Version: 6 Date: 7 December 2008

Author's response to reviews:

Spelling and some vocabulary corrected.